DATE LECTURE TOPIC HOMEWORK DUE Aug 26 Introduction and Describing Motion Ch. 2, Ex. 10, 12, 17, 22 Sep 4 Aug 28 Describing Motion Ch. 2, Ex. 27, 28, 33, 34, Sep 4 Sep 2 Labor Day Holiday Sep 4 Falling Bodies Ch. 2, Ex. 52, 68, 70, 76, 80 Sep 11 Sep 9 Newton’s Law of Motion I Ch. 4, Ex. 3, 18, 23, 36 Sep 18 Sep 11 Newton’s Law of Motion II Ch. 4, Ex. 45, 60, 99 Sep 18 Sep 16 Projectile Motion Ch. 3, Ex. 50, 54, 59, 70 Sep 25 Sep 18 Circular Motion Ch. 7, Ex. 7, 9, 11, 26, 29 Sep 25 Sep 23 Gravitation Ch. 7, Ex. 4, 65, 71, 74 Oct 2 Sep 25 Test 1. (Ch 2-4, parts of 7) Sep 30 Test 1 Explanation Oct 2 First Look at Momentum and Its Conservation Ch. 6, Ex. 8, 27, 33, 35 Oct 9 Oct 7 Second Look at Momentum and Its Conservation Ch. 6, Ex. 39, 53, 60, 67 Oct 16 Oct 9 Intro to Work and Energy Ch. 5, Ex. 6, 7, 12, 34, 45 Oct. 16 Oct 14 Yet More Work and Energy Ch. 5, Ex. 55, 62, 74, 79 Oct 23 Oct 16 Bodies in Equilibrium Ch. 8, Ex. 19, 20, 21, 28 Oct 23 Oct 21 Rotational Motion Ch. 8, Ex. 41, 49, 58, 72 Oct 30 Oct 23 Angular Momentum Ch. 8, Ex. 80, 89, 90, 92 Oct 30 Oct 28 Static Fluids and Why Things Float Ch. 9, Ex. 24, 26, 36, 53, 58 Nov 6 Oct 30 Test 2 (Ch 5-6, parts of 7, 8-9) Nov 6 Temperature and Ideal Gas Behavior Ch. 10, Ex. 9, 12, 31, 33, 34, Nov 13 Nov 11 First Look at Simple Harmonic Motion Ch. 13, Ex. 6, 10, 14, 16 Nov 20 Nov 13 Last Look at Simple Harmonic Motion Ch. 13, Ex. 20, 24, 29, 31, 38 Nov 20 Nov 18 Wave Motion and Sound Ch. 13, Ex. 42, 46, 49, 51, 64, 68 Nov 27 Nov 27 Test 3 (Ch. 10-11, 13-14)

Final Examination: 123 FS 11:30 a.m. { 2:30 p.m. Tuesday, 10 December 1996